\documentclass[12pt,fleqn]{article}

\setlength{\topmargin}{-1.3in}
\setlength{\textheight}{10.5in}
\setlength{\evensidemargin}{-.2in}
\setlength{\oddsidemargin}{-.2in}
\setlength{\textwidth}{6.5in}
\renewcommand{\baselinestretch}{1.2}

%length{\topmargin}{-.5in}
%\setlength{\textheight}{0.9in}
%\setlength{\textwidth}{6.5in}
%\setlength{\oddsidemargin}{-.2in}
%\setlength{\evensidemargin}{-.5in}

\newlength{\newitemsep}
\setlength{\newitemsep}{-5pt plus 3pt}

\pagestyle{empty}
\begin{document}

%\Large
\begin{center}
{\bf Math 1090 - 2

Syllabus -- Summer 1999}
\end{center}
%\normalsize

\smallskip

\begin{description}

\item[{\bf Instructor:}]  Blake Thornton, Building 122 (room \#102), 581-4027

\item[{\bf Email:}]  thornton@math.utah.edu

\item[{\bf Office Hours:}] After class

\item[{\bf Text:}]  Mathematical Applications, we will cover chapters 1-6.

\item[{\bf Prerequisites:}] Math 1010, Chapter 0 of text.
Are you in the right class?

\item[{\bf Course Description:}] Chapters 1-6 of text.

\item[{\bf Calculator:}] 
You will need a calculator with at least exponential, logarithm, and $y^x$ 
functions.
Graphing and financial calculators are nice, but not necessary.
You are responsible for know how your calculator works.

\item[{\bf Tutoring:}] 
There is free tutoring available in rooms 241 \& 259 of building 129
which is open the following times:

\begin{tabular}{ll}
Monday: & 9AM-4:30PM \\
Tuesday: & 9AM-2PM \& 5PM-7PM \\
Wednesday: & 9AM-7PM \\
Thurs/Fri: & 9AM-2PM 
\end{tabular}


\item[{\bf Homework:}] 
Homework will not be collected, but it must be completed daily in order
to learn the material.
Homework is assigned for every section.
You should consider this as a suggested minimum number of problems to do,
you may need to do more.

\item[{\bf Grades/Tests/Quizzes:}]  
There will be 3 in class exams, a bunch of quizzes and a comprehensive final.
The tests will emphasize the material covered since the last test, 
but anything we have covered is fair game.  
We will usually have a \underline{quick} review near the end of the class the meeting
before a test.
We will have quizzes on many of the days that there is no test.

\[
\begin{array}{ll}
%
15\% &
\begin{array}{ll}
\mbox{{\bf Quizzes}} & \mbox{Days there are not tests} 
\end{array} \\
%
55\%  & \left\{ 
\begin{array}{lll}
\mbox{{\bf Test 1}} & \mbox{Monday} & \mbox{May 24} \\
\mbox{{\bf Test 2}} & \mbox{Friday} & \mbox{June 4} \\
\mbox{{\bf Test 3}} & \mbox{Monday} & \mbox{June 14} \\
\end{array}
\right. \\
%
30\% &
\begin{array}{llll}
\mbox{{\bf FINAL EXAM}} & \mbox{Wednesday} & \mbox{June 23} & \mbox{9:00-11:00}
\end{array}
%
\end{array}
\]

\end{description}



\end{document}




